Author’s response to reviews

Title: A review of clinical trial designs used to detect a disease-modifying effect of drug therapy in Alzheimer's disease and Parkinson's disease

Authors:
David McGhee (mailto:davidjmmcghee@gmail.com; mailto:david.mcghee@abdn.ac.uk)
Craig Ritchie (mailto:craig.ritchie@ed.ac.uk)
John Zajicek (jz45@st-andrew.ac.uk)
Carl Counsell (carl.counsell@abdn.ac.uk)

Version: 3 Date: 10 Apr 2016

Author’s response to reviews:

Editorial Office
BMC Neurology
BioMed Central
236 Gray’s Inn Road
London
WC1X 8HB

10th April 2016

Dear Sir/Madam

A review of clinical trial designs used to detect a disease-modifying effect of drug therapy in Alzheimer’s disease and Parkinson’s disease.

We are grateful for your consideration of our manuscript for inclusion in BMC Neurology. We have amended the title as requested.

Thank you again for considering our article.

Yours sincerely
David McGhee
Corresponding Author
Honorary Clinical Research Fellow
University of Aberdeen
mailto:davidjmmcghee@gmail.com